Author's response to reviews

Title: Effectiveness of anodal transcranial direct current stimulation in patients with chronic low back pain: Design, method and protocol for a randomised controlled trial.

Authors:

Kerstin Luedtke (kerstin.luedtke@googlemail.com)
Alison Rushton (a.b.rushton@bham.ac.uk)
Christine Wright (c.c.wright@bham.ac.uk)
Tim P Juergens (t.juergens@uke.uni-hamburg.de)
Gerd Mueller (g.mueller@ruecken-zentrum.de)
Arne May (a.may@uke.uni-hamburg.de)

Version: 2 Date: 20 December 2011

Author's response to reviews:

The abstract has been changed to the suggested format using the subheadings background, methods/design, discussion.

A functioning link to reference 50 was provided.

The figure has been cropped as closely as possible to minimise white space around the image.

Typography has been checked for spelling mistakes.